CrossMark

Citation: Jenjaroen K, Chumseng S, Sumonwiriya M, Ariyaprasert P, Chantratita N, Sunyakumthorn P, et al. (2015) T-Cell Responses Are Associated with Survival in Acute Melioidosis Patients. PLoS Negl Trop Dis 9(10): e0004152. doi:10.1371/journal. pntd.0004152

Editor: Joseph M. Vinetz, University of California San Diego School of Medicine, UNITED STATES

Received: June 22, 2015

Accepted: September 18, 2015

Published: October 23, 2015

Copyright: © 2015 Jenjaroen et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: SJD is grateful for the support of a Wellcome Trust Intermediate Clinical Fellowship award ref WT100174/Z/12/Z, an Academy of Medical Sciences Starter Grant for Clinical Fellows (2011-2013) and an Oxford University Clinical Academic Graduate School Academic Lectureship (2009-2013). Mahidol-Oxford Tropical Medicine Research Unit is supported by the Wellcome Trust. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

T-Cell Responses Are Associated with Survival in Acute Melioidosis Patients

Kemajittra Jenjaroen¹, Suchintana Chumseng¹, Manutsanun Sumonwiriya¹, Pitchayanant Ariyaprasert¹, Narisara Chantratita^{1,2}, Piyanate Sunyakumthorn¹, Maliwan Hongsuwan¹, Vanaporn Wuthiekanun¹, Helen A. Fletcher³, Prapit Teparrukkul⁴, Direk Limmathurotsakul^{1,5}, Nicholas P. J. Day^{1,6}, Susanna J. Dunachie^{1,6}*

 Mahidol-Oxford Tropical Medicine Research Unit, Mahidol University, Bangkok, Thailand, 2 Department of Microbiology and Immunology, Faculty of Tropical Medicine, Mahidol University, Bangkok, Thailand,
London School of Hygiene and Tropical Medicine, London, United Kingdom, 4 Sappasithiprasong Hospital, Ubon Ratchathani, Thailand, 5 Department of Tropical Hygiene, Faculty of Tropical Medicine, Mahidol University, Bangkok, Thailand, 6 Centre for Tropical Medicine, University of Oxford, Oxford, United Kingdom

* susie.dunachie@ndm.ox.ac.uk

Abstract

Background

Melioidosis is an increasingly recognised cause of sepsis and death across South East Asia and Northern Australia, caused by the bacterium *Burkholderia pseudomallei*. Risk factors include diabetes, alcoholism and renal disease, and a vaccine targeting at-risk populations is urgently required. A better understanding of the protective immune response in naturally infected patients is essential for vaccine design.

Methods

We conducted a longitudinal clinical and immunological study of 200 patients with melioidosis on admission, 12 weeks (n = 113) and 52 weeks (n = 65) later. Responses to whole killed *B. pseudomallei* were measured in peripheral blood mononuclear cells (PBMC) by interferon-gamma (IFN- γ) ELIspot assay and flow cytometry and compared to those of control subjects in the region with diabetes (n = 45) and without diabetes (n = 43).

Results

We demonstrated strong CD4+ and CD8+ responses to *B. pseudomallei* during acute disease, 12 weeks and 52 weeks later. 28-day mortality was 26% for melioidosis patients, and *B. pseudomallei*-specific cellular responses in fatal cases (mean 98 IFN- γ cells per million PBMC) were significantly lower than those in the survivors (mean 142 IFN- γ cells per million PBMC) in a multivariable logistic regression model (*P* = 0.01). A J-shaped curve association between circulating neutrophil count and mortality was seen with an optimal count of 4000 to 8000 neutrophils/µl.

Competing Interests: The authors have declared that no competing interests exist.

Melioidosis patients with known diabetes had poor diabetic control (median glycated haemoglobin HbA_{1c} 10.2%, interquartile range 9.2–13.1) and showed a stunted *B. pseudo-mallei*-specific cellular response during acute illness compared to those without diabetes.

Conclusions

The results demonstrate the role of both CD4+ and CD8+ T-cells in protection against melioidosis, and an interaction between diabetes and cellular responses. This supports development of vaccine strategies that induce strong T-cell responses for the control of intracellular pathogens such as *B. pseudomallei*.

Author Summary

Melioidosis is a key cause of death in South East Asia and Northern Australia. It is caused by the soil-dwelling bacteria *Burkholderia pseudomallei*, and presents as a range of clinical illnesses including pneumonia and bloodstream infections. About two-thirds of patients with melioidosis in Thailand have diabetes, but the immune responses associated with death and diabetes are unknown. This study examined the relationship between immune responses to the bacteria and death by studying the bacteria-specific lymphocyte responses in 200 patients admitted to hospital with acute melioidosis and following the patients for up to one year where possible. 26% of patients died within 28 days despite receiving antibiotics and supportive care. We showed people with melioidosis make bacteria-specific lymphocyte responses, with lower levels seen in fatal cases compared to survivors. People with diabetes make lower responses than people without diabetes, and higher levels of circulating neutrophils on admission to hospital were associated with lower lymphocyte responses both during illness and three months later in survivors. This highlighting of lymphocyte responses to melioidosis is important for the design of vaccines to target at risk groups.

Introduction

Melioidosis is a severe and often fatal disease common in Southeast Asia and Northern Australia, caused by the soil-dwelling bacterium *Burkholderia pseudomallei*. The main routes of infection are skin inoculation, inhalation, and ingestion of the organism from the environment [1]. The known risk factors are diabetes, alcoholism, renal disease and increasing age, with HIV not a major risk factor [2]. There is a range of presentations including pneumonia, liver and splenic abscesses and septic shock, and the mortality in North-East Thailand is around 40% despite appropriate antibiotic therapy [3]. Melioidosis is now recognised as endemic in an expanding number of countries [4]. The prevalence of diabetes in the region is growing, with the reported Thai prevalence now 6.9% (chiefly Type 2) [5]. Meanwhile changing demographics in the region leave an increasingly ageing population exposed to the bacterium by farming as younger family members migrate to urban regions [6]. A vaccine is therefore in demand and cost effective if targeted towards at-risk groups [7].

The human immune response to melioidosis is far from fully understood. Important roles for Toll-like receptor (TLR) [$\underline{8-10}$], and nucleotide-binding oligomerization domain 2 (NOD2) [$\underline{11}$] variants, macrophage function [$\underline{12}$] and neutrophil response in diabetic patients to *B. pseudomallei* [$\underline{13,14}$] have been reported. Up to 80% of people in endemic regions have serological responses to *B. pseudomallei* whether they have a history of disease or not [$\underline{15,16}$].

As an intracellular pathogen [17] capable of chronic disease and latency, cellular adaptive immunity seems likely to be important. There are similarities in the host response to *B. pseudomallei* and to *M. tuberculosis*, with both being intracellular, sharing the risk factor of diabetes [18] and showing similar gene expression profiles dominated by interferon-mediated signalling pathways [19]. Further support for cellular immunity comes from reports of individuals with HLA-DRB1*1602 in Thailand being at increased risk of melioidosis [20]. Murine models have demonstrated the importance of IFN- γ in survival from experimental challenge [21] and production of IFN- γ in response to *B. pseudomallei* by NK (Natural Killer) cells, CD8+ T-cells and CD4+ T-cells [22]. In humans, an association between survival from melioidosis and IFN- γ levels and linked cytokines has been shown [23–25]. Cell mediated immunity has been demonstrated in Australian patients recovered from melioidosis [26]. Memory T-cell responses to *B. pseudomallei* in healthy blood donors in an endemic region and some subjects with previous melioidosis have been characterised [27]. However the relationship between cellular responses and survival has not previously been characterised during acute melioidosis.

Vaccine design requires knowledge of naturally acquired immunity, to understand which bacterial antigens are immunogenic and which components of the immune response are protective. The development of vaccines against melioidosis is hindered by knowledge gaps [28]. The aim of this study was to see if the cellular responses to *B. pseudomallei* were related to survival in patients with melioidosis, and to see the dynamics of the cellular response over one year in survivors from melioidosis. This work demonstrates an association between lower CD4 + and CD8+ T-cell responses and mortality from the disease, and shows impaired cellular immunity in diabetic patients acutely unwell with melioidosis compared to non-diabetic patients. These findings support an important contribution to control of melioidosis in humans by cellular immunity.

Materials and Methods

Ethical approval

The study was approved by the ethics committees of Faculty of Tropical Medicine, Mahidol University (Submission number TMEC 12–014), of Sappasitthiprasong Hospital, Ubon Ratchathani (reference 018/2555) and the Oxford Tropical Research Ethics Committee (reference 64–11). The study was conducted according to the principles of the Declaration of Helsinki (2008) and the International Conference on Harmonization (ICH) Good Clinical Practice (GCP) guidelines. Written informed consent was obtained for all patients enrolled in the study.

Recruitment of subjects

In-patients at Sappasitthiprasong Hospital, Ubon Ratchathani over 18 years of age with melioidosis were recruited (*Melioid Cohort*) following positive culture of *B. pseudomallei* from a clinical specimen, alongside healthy control subjects attending the hospital's blood donation clinic (*Healthy Control Cohort*) and subjects attending the hospital's diabetes outpatient clinic (*Diabetes Control Cohort*). For patients who did not attend follow-up, their 28 day survival status was determined by using the hospital mortality records and by telephone.

Blood samples

Subjects enrolled in the study had 25ml of blood drawn for isolation of peripheral blood mononuclear cells (PBMC), serum and glycated haemoglobin (HbA_{1c}) testing for the *Melioid* and *Diabetes Cohorts*. PBMC and plasma were separated by density centrifugation within 3 hours of blood draw and counted with a *Scepter* handheld counter (*Millipore*, *UK*) before use in fresh assays.

Antigens for assays

Whole heat-inactivated *B. pseudomallei* (HIA-Bp) was prepared from two Thai patient isolates 199a and 207a as previously described [29] and used at a dilution of 1:60 (20 μ g/ml by Bicinchoninic Acid Assay, *Sigma*) in the ELIspot assay. Phytohemagglutinin (PHA) (final concentration 5 μ g/mL) was added to positive control wells. A T-cell epitope pool (*Mabtech*, *AB*, *Sweden*, final concentration 1 μ g/mL) was used as control antigens.

Ex-vivo interferon- γ (IFN- γ) Enzyme-linked immunosorbent spot-forming cell assay (ELIspot) and Indirect Haemagglutination Assay (IHA)

The kinetics and magnitude of the cellular response to whole killed *B. pseudomallei* were assessed by fresh *ex-vivo* IFN- γ ELIspot assay. Briefly, 96-well Multiscreen-I plates (*Millipore*, UK) were coated overnight with 1D1K anti-human IFN- γ (*Mabtech, AB, Sweden*) at 4°C. Fresh PBMCs were added in duplicate wells at 2x10⁵ PBMCs/well and each antigen was added at the optimal concentration. After 18 hours, secreted IFN- γ was detected according to the manufacturer's instructions (*Mabtech, AB, Sweden*) and read using a CTL ELIspot reader. Results are expressed as IFN- γ spot-forming cells (SFC) per million PBMC. Background responses in unstimulated control wells were typically less than 20 spots, and were subtracted from those measured in peptide-stimulated wells. IHA was performed as described previously [29] and a titre of 1:40 or greater was considered positive [30].

Intracellular staining cytokine assay by flow cytometry

PBMC were stimulated for 18 hours with heat-inactivated *B. pseudomallei* (50µg/well) or media. Brefeldin A (Ebioscience, USA) was added at 10 ug/ml and following 4 hours of further incubation, staining for intracellular APC-IFN- γ and for the immune cell surface markers PCP-anti-CD3, FITC-anti-CD4, APCH7-anti-CD8, PE-anti-CD56 and V450-CD14 all BD Biosciences, USA) was performed. Samples were analysed using a MACSQuant Analyzer 10 (Miltenyi Biotec, Germany) with Flowjo software (Treestar Inc, USA). The gating strategy used to identify the cellular source of IFN- γ after stimulation with heat-killed *B. pseudomallei* was firstly selection of singlets to exclude aggregates. The lymphocyte populations were then selected for immunophenotyping: double positive CD3+CD4+ for CD4 T cells, double positive CD3+CD8+ for CD8 T cells, double positive, cells were gated from singlets. IFN- γ expression levels were then determined by analysis of each cell phenotype. Responses were quantitated for the negative control-subtracted percentage of IFN- γ secreted CD4+ or CD8+ cells from the gated lymphocyte population.

Data management and statistical analysis

Study clinical data were entered into a password-protected web-based database (*OpenClinica* version 3.1, *LLC and collaborators, Waltham, MA, USA*). The presence of diabetes mellitus in *Melioid Cohort* subjects was defined for analysis as previous diagnosis by a doctor and / or having an HbA_{1c} of \geq 7% [31]. Subjects in the *Healthy Control Cohort* and *Diabetes Control Cohort* were excluded from the analysis if their IHA titre was greater or equal to 1:40, to allow these cohorts to serve as negative controls. Statistical analysis was performed using Graphpad Prism version 6 (*San Diego, CA, USA*) and IBM SPSS Statistics for Windows version 22.0

(*Armonk*, *NY*, *USA*). Non-parametric tests were used (Mann Whitney, paired Wilcoxon, Kruskal-Wallis and Spearman's rank correlation tests) and groups were compared with 2-tailed Fisher's exact test. To examine the relationship between mortality and the *B. pseudomallei*-specific cellular response by ELISpot, a multivariable logistic regression modelling was developed using a purposeful selection method [32] with logarithmic transformation of non-parametric data. A *P* value (2-tailed) of <0.05 was considered significant.

Results

Subjects

200 patients with culture-confirmed acute melioidosis were recruited from September 2012 until September 2014 into the *Melioid Cohort* and followed up at 12 weeks (n = 113, 57%) and 52 weeks (n = 94, 47%, follow-up ongoing). On average, participants were enrolled on day 5 after admission (median 5, interquartile range IQR 3 to 6 days). The demographics of the subjects in the study are shown in Table 1. The clinical site of culture of *B. pseudomallei* for the *Melioid Cohort* was blood for 105/200 (53%) of cases, with no significant difference between those with and without diabetes. 172/200 (86%) had at least one identifiable risk factor such as diabetes, renal or heart disease, alcohol excess or age greater than 65 years. Antibody titres to *B. pseudomallei* were greater than or equal to 40 by indirect haemagglutination assay in 75% (95% CI 68–81) of acute melioidosis patients, consistent with previous reports [1]. 50 subjects were recruited from diabetes outpatients for the *Diabetes Control Cohort*, and 5 were excluded from the analysis for having an IHA ≥40. 50 subjects were recruited from the blood donation clinic for the *Healthy Control Cohort*, and 7 were excluded from the analysis for having an IHA ≥40.

Diabetes

117 subjects with acute melioidosis were known to have diabetes on admission, with a mean HbA_{1c} of 11.1% (median 11.3, IQR 9.1–13.1). The mean HbA_{1c} of the out-patient diabetic control group was lower at 8.2% (median 7.8, IQR 7.2–9.1) (P < 0.001, Fig 1A). Drug information was available for 108/117 of subjects admitted with known diabetes and is shown in Fig 1B. 43 subjects (37%) received at least one oral hypoglycaemic drug including the sulphonylurea glipizide (n = 26), metformin (n = 21), glyburide (n = 4), pioglitazone (n = 4) and glimepiride (n = 1). In addition 17 new diagnoses of diabetes were made on the basis of HbA_{1c} in patients recruited without a previous history of diabetes.

Table 1. Subject demographics.

Cohort	N	Gender		Age In years Mean (range)	Diabetes*	
		Male n (%)	Female n (%)		Pre-diagnosed* n (%)	Study definition* n (%)
Melioid Cohort all	200	133 (67%)	67 (33%)	55 (19–89)	117 (59%)	134 (67%)
Melioid Cohort with Diabetes	134	84 (63%)	50 (37%)	53 (19–89)	117 (87%)	134 (100%)
Melioid Cohort no Diabetes	66	49 (74%)	17 (26%)	58 (25–88)	0 (0%)	0 (0%)
Diabetes Controls Cohort	45	30 (67%)	15 (33%)	57 (23–76)	45 (100%)	45 (100%)
Heathy Controls Cohort	43	26 (60%)	17 (40%)	48 (33–73)	0 (0%)	0 (0%)

The gender, age and diabetes status of subjects in the study are shown. Two thirds (67%) of the patients with acute melioidosis were male, with a mean age of 55 years.

*Diabetes definitions are as follows: Pre-diagnosed = subjects with a previous diagnosis of diabetes mellitus by a doctor. Study definition = subjects with a previous diagnosis of diabetes mellitus by a doctor and / or having an HbA_{1c} of \geq 7%.

doi:10.1371/journal.pntd.0004152.t001



(b) DRUG TREATMENT OF PATIENTS ADMITTED WITH MELIOIDOSIS AND KNOWN DIABETES MELLITUS



Fig 1. HbA_{1c} for Melioid and Diabetes Cohorts, and treatment received. Fig 1A shows boxplots with Tukey whiskers showing the percentage glycated haemoglobin (HbA_{1c}) for subjects in the *Melioid Cohort* (culture-confirmed acute melioidosis) at admission to the study with no diagnosis of Diabetes Mellitus (*Melioid & no DM*), with a new diagnosis of Diabetes Mellitus defined for the study as an HbA_{1c} value of \geq 7% (*Melioid & New Diagnosis DM*) and with known Diabetes Mellitus (*Melioid & Known DM*) alongside subjects recruited from diabetes out-patient clinic who are (*DM outpatient controls*).**** *P* < 0.0001 by Mann-Whitney test. Fig 1B shows the drug treatment for diabetes on admission for the 117 patients in the *Melioid Cohort* with previously diagnosed diabetes mellitus.

doi:10.1371/journal.pntd.0004152.g001

Clinical outcomes

Mortality data was available for 198/200 patients in the Melioid Cohort. 51/198 patients (26%) died within 28 days of admission to hospital despite correct diagnosis and appropriate choice of antibiotic. A further 13 patients died after 28 days (from 29 to 90 days). 28 day mortality was 24% (32/133; 95% CI 17-32) for those with diabetes and 29% (19/65; 95% CI 19-42) for those without diabetes, P = 0.49 by Fisher's exact test). 15/19 (79%) of the non-diabetic subjects who died had at least one identifiable risk factor such as liver or renal disease, excessive use of alcohol or age greater than 65 years. Increasing age was a risk factor for mortality in univariable analysis (P = 0.009) but was not an independent risk factor (P = 0.29) in the multivariable logistic regression model. Increasing IHA titres were associated with survival (P = 0.002) in the univariable analysis, but this was not significant by multivariable analysis and did not contribute to the logistic regression model. Bacteraemia was closely associated with mortality compared to no bacteraemia (P < 0.0001) with 39/51 (76%) of the deaths occurring in people who were blood culture positive. The median length of time from admission to hospital until death was 10 days (IQR 5–17), with 35/51 patients taken from hospital to die at home with family. There was no significant relationship between sulphonylurea use in the diabetic melioidosis patients and mortality (P = 0.65). Four surviving patients (all with diabetes) had a subsequent diagnosis of melioidosis 2, 5, 6 and 20 months after study entry, defined by positive culture for B. pseudomallei preceded by negative cultures following standard care including 12 weeks of planned oral follow-on therapy with co-trimoxazole.

Patients with melioidosis have durable specific T-cell responses to *B*. *pseudomallei*

Cellular immunity to *B. pseudomallei* was demonstrated in patients with acute melioidosis (Fig 2A). The mean IFN- γ ELIspot response was 133 SFC per million PBMC (median 25, IQR 3 to 99) at Week 0, 262 (median 70, IQR 15 to 265) at Week 12 and 164 (median 30, IQR 10 to 74) at Week 52. These responses remained elevated at Week 52 compared to the diabetic outpatient controls (mean 37, median 5, IQR 1 to 26, *P* = 0.004) and healthy seronegative controls (mean 15, median 5, IQR 1–23, *P* = 0.0002), whose responses did not differ significantly from each other (*P* = 0.99). Responses to control antigens and media are shown in Fig 2B. Phenotypic analysis demonstrated IFN- γ production by a range of cell phenotypes including CD4, CD8, NK and CD14, with the percentage of IFN- γ producing CD8 cells increasingly significantly from Week 0 to Week 52 (*P* = 0.01, Fig 2C).

Depressed T-cell responses to B. pseudomallei at admission were associated with higher mortality rates

The mean cellular response in subjects in the *Melioid Cohort* who survived the illness was 142 SFC per million PBMC (median 35 IQR 1–145) compared to 98 (median 3 IQR 1–43) for those who died (Fig 3A). For other antigens tested, there was no significant difference between survivors and non survivors.

The logistic regression model measured the effect of cellular responses to *B. pseudomallei*, neutrophil count, age, diabetes status, and the risk factor of known renal disease on the likelihood that participants died within 28 days of admission with acute melioidosis (Table 2). Log_{10} transformed data was used for non-parametric distributions. A tenfold increase in *B. pseudomallei*-specific cellular response was associated with reduction by almost half of the likelihood of death (OR 0.56, 95% CI 0.36 to 0.88, *P* = 0.012). Survivors showed an increased frequency of both CD4+ IFN- γ T-cells (*P* = 0.04) and CD8+ IFN- γ T-cells (*P* = 0.009) compared to non-



T-cell responses to Heat-killed whole Burkholderia pseudomallei

Fig 2. T-cell Responses to Heat killed B. pseudomallei (Fig 2A) to control antigens (Fig 2B) and by cell phenotype (Fig 2C). Peripheral blood mononuclear cells (PBMC) from patients with acute melioidosis (*Melioid Cohort Wk0*) and 12 (*Wk12*) and 52 (*Wk 52*) weeks later were tested alongside PBMC from diabetic outpatients (*Diabetes*) and healthy endemic seronegative control subjects (*Healthy*). Cells were stimulated with heat-inactivated *B. pseudomallei* (Fig 2A), a control antigen T-cell epitope pool or media only (Fig 2B) for 18 hours and IFN-γ secreting cells counted and expressed as spot forming cells per million PBMC (SFC/10⁶ PBMC). To characterise the cell phenotype producing IFN-γ, PBMC were incubated with whole *B. pseudomallei* for 18 hours and stained for intracellular IFN-γ versus immune cell surface markers (PCP-anti-CD3, FITC-anti-CD4, APCH7-anti-CD56 and V450-CD14, Fig 2C). Horizontal lines represent medians, ***P<0.001, **P<0.01, *P<0.05, ns = not significant, testing by Kruskal-Wallis with Dunn's correction for multiple testing.

doi:10.1371/journal.pntd.0004152.g002

survivors (Fig 3B). The IFN- γ + CD8+ fraction was 0.05% or less in 14/14 fatal cases studied compared to 5/16 of survivor cases.

Subjects who were bacteraemic had lower cellular responses than subjects who were blood culture negative (P = 0.007). The mean cellular response in subjects in the *Melioid Cohort* who were bacteraemic was 99 SFC per million PBMC (median 18 IQR 1–70) compared to 170 (median 46 IQR 5–171) for those who were not bacteraemic.

In the acute stages of melioidosis depressed cellular responses to B. pseudomallei are observed in patients with diabetes

The cellular response in subjects with acute melioidosis and diabetes was lower than in subjects with acute melioidosis and no diabetes (mean 101 SFC per million PBMC, median 20, IQR 3–76, compared to mean 198, median 49, IQR 6–196), both at Week 0 (P = 0.03, Fig 4). There was a non-significant trend for lower responses in subjects with diabetes 12 and 52 weeks later, when subject numbers were lower. There was a non-significant trend towards higher HbA_{1c} being associated with lower cellular responses at Week 0 (Spearman's r = -0.133, P = 0.07) with no clear trend at subsequent timepoints.

High neutrophil counts during acute melioidosis are associated with lower B. pseudomallei-specific cellular responses in recovery

A higher neutrophil count on admission was also associated with death (median 12,217 neutrophils / μ l in fatal cases versus 9,495 neutrophils / μ l in survived cases, *P* = 0.002 on univariable analysis). In the multivariable model (Table 2) a "J-shaped curve" effect was seen with higher mortality for both low neutrophil counts (adjusted OR 5.8 [1.1–29.6] for < 4000 neutrophils / μ l compared to > 4000 to 8000) and for higher neutrophil counts (adjusted OR 18.9 [3.1–114.2] for > 20,000 neutrophils / μ l compared to > 4000 to 8000).

The cellular response to *B. pseudomallei* 12 weeks after the illness showed a tight negative association with the peripheral neutrophil count on admission to hospital (P < 0.0001, <u>S1 Fig</u>). This relationship between neutrophil count and cellular response was also present at Week 0 (P = 0.001), and Week 52 (P = 0.008). No such relationship was seen for the control antigens studied.

Discussion

The results of this study demonstrate a high mortality for patients with melioidosis in Ubon Ratchathani as previously reported [3]. Work continues in the hospital to audit and improve the "door-to-needle time" for sepsis patients receiving antibiotics but it should be noted that not all patients culture positive for *B. pseudomallei* present to hospital with a classical acute sepsis syndrome. The mortality predominantly occurred in patients with bacteraemia rather





Fig 3. Ex vivo interferon-gamma ELISPOT responses by survival. Peripheral blood mononuclear cells (PBMC) from patients with acute melioidosis at study entry (Week 0) were stimulated with heat-inactivated *B. pseudomallei* (Bp), or a control antigen panel for 18 hours and IFN-γ secreting cells counted and expressed as spot forming cells per million (SFC/10⁶) PBMC. Responses were compared for survivors and those who died (Fig 3A). To characterise the cell phenotype producing IFN-γ, PBMC were incubated with whole *B. pseudomallei* for 18 hours and stained for intracellular IFN-γ versus immune cell surface markers (PCP-anti-CD3, FITC-anti-CD4 and APCH7-anti-CD8, Fig 3B). Horizontal lines represent medians, ****P*<0.001, ***P*<0.05, ns = not significant, by logistic regression adjusted for age, diabetes status, renal disease and neutrophil count for responses in Fig 3A and by Mann-Whitney test for Fig 3B.

doi:10.1371/journal.pntd.0004152.g003

than contained disease, as previously described [33], with no significant difference in mortality between subjects with diabetes versus no diabetes. Increasing age was a risk factor for death on univariable analysis as previously reported [34]. Age was not an independent risk factor in the multivariable model where cellular immunity and neutrophil count were independent predictors of mortality and may represent age-related immune mechanisms of decline.

Lower *B. pseudomallei*-specific T-cell responses were associated with increased mortality. Both CD4+ and CD8+ IFN- γ -producing cells were associated with survival. CD4+ cells, but not CD8+ cells have previously been shown by depletion studies to be important in mediating immunity against experimental melioidosis in a murine model [35].The additional importance of CD8 T-cells in control of melioidosis in humans is supported by our finding that the IFN- γ CD8+ fraction was very low (0.05% or less) in every fatal case whilst the majority of survivors had higher responses. CD8+ cells have been shown to be a source of IFN- γ during primary infection of naïve mice [36] or cell lines [37] via bystander activation but the expansion of *B. pseudomallei*-specific IFN- γ CD8+ cell fraction at one year compared to during acute illness supports an adaptive immune role.

The reduced cellular response seen in fatal cases could reflect general immune failure as part of multi-organ failure in the terminal stages. However no such relationship was seen for patients in this study with cellular responses to a control T-cell epitope pool. Establishing the cellular response as having a causal direct role in preventing death, rather than being a marker of outcome is difficult in patient studies, but the correlation demonstrated here supports the hypothesis that T-cell responses are important in the control of *B. pseudomallei* infection in

Variable		Adjusted Odds Ratio (95% CI)	P value
Age		1.0 (0.99–1.1)	0.29
Diabetes		0.56 (0.24–1.3)	0.2
Pre-existing renal disease		3.9 (1.5–10.5)	0.007
Bp cell response		0.56 (0.36–0.88)	0.01
Neutrophil count / µl	> 4000–8000	1	1
	\leq 4000	5.8 (1.1–29.6)	0.04
	> 8000–12,000	8.1 (2.1–31.5)	0.002
	> 12,000–20,000	7.5 (1.8–30.6)	0.005
	> 20,000	18.9 (3.1–114.2)	0.001

Table 2. Multivariable analysis for prediction of 28-day mortality in patients with acute melioidosis.

The table displays the odds ratios for variables studied as predictors of mortality in 200 patients with acute melioidosis in a multivariable logistic regression model. Bp cell response is the log_{10} transformed ELIspot response in fresh peripheral blood mononuclear cells to heat-killed *B. pseudomallei*. Neutrophil count represents the peripheral blood neutrophil count with > 4000 to 8000 neutrophils / µI as the comparator group. Odds ratios are adjusted for the other variables included in the model.

doi:10.1371/journal.pntd.0004152.t002



Fig 4. Ex vivo interferon-gamma ELISPOT Responses to Heat killed Burkholderia pseudomallei by diabetes status. Ex vivo IFN- γ ELIspot responses to heat-inactivated *B. pseudomallei* are shown for subjects with melioidosis on admission (*Week 0*) and 12 (*Week12*) and 52 (*Week 52*) weeks later according to the presence (+) or absence (-) of a diagnosis of Diabetes Mellitus. Peripheral blood mononuclear cells (PBMC) were stimulated with heat-inactivated *B. pseudomallei* for 18 hours and IFN- γ secreting cells counted and expressed as spot forming cells per million PBMC (SFC/10⁶ PBMC). Subjects with diabetes had lower responses to *B. pseudomallei* during acute illness (Week 0) compared to subjects with no diabetes. Boxplots with Tukey whiskers are shown with responses greater than 1000 SFC/million PBMC not displayed. ** *P* < 0.01, ns = not significant by Kruskal-Wallis testing with Dunn's multiple comparisons test.

doi:10.1371/journal.pntd.0004152.g004

humans. Further depletion and adoptive transfer studies in animal models are required to establish causality.

This is the first study to demonstrate the dynamics of the *B. pseudomallei*-specific cellular response from acute melioidosis through to follow-up one year later, and the durability of the IFN- γ response is shown. The IFN- γ responses demonstrated in survivors 12 weeks after admission with acute melioidosis are compatible with responses reported in a previous study of T-cell responses to *B. pseudomallei* using a longer (42 hour) incubation protocol [27]. The source of the IFN- γ is a combination CD4+ and CD8+ T-cells and Natural Killer cells as previously shown [27] as well as some monocyte production, and we have shown an increase in the

B. pseudomallei-specific IFN- γ CD8+ cell fraction at one year compared to during acute illness. Ongoing studies are evaluating the dynamics of the multifunctional, memory and regulatory cell subsets over time.

The finding of a "J-shaped curve" association between mortality and peripheral blood neutrophil count during acute melioidosis is compatible with the existing literature on the relationship between neutrophils and sepsis [38]. A count in the range of 4000 to 8000 neutrophils/ μ l was associated with the lowest mortality, and patients with counts above 20,000 neutrophils/ μ l had an adjusted odd ratio for death by 28 days of 18-fold. Neutrophils play a key role in the host response to *B. pseudomallei* [13,39,40] bacteria, but excess activation and inappropriate distribution of neutrophils have been linked to multi-organ failure and poor outcomes in sepsis [38].

There was an inverse relationship between neutrophil count on admission to hospital and the cellular response to *B. pseudomallei* on admission, 12 weeks later and 52 weeks later for survivors. Excess neutrophil activation may represent a shift in the development of hematopoietic stem cell progenitors in the bone marrow from lymphoid lineage to myeloid lineage. Although neutrophils are generally believed to support antigen presentation and the development of specific adaptive responses through multiple pro-inflammatory processes, the relationship is complex. Studies have shown inhibition of T-cell responses by neutrophils in bacterial sepsis [41] and by related immature myeloid cells named myeloid derived suppressor cells (MDSCs) [42]. A recent elegant study demonstrated that CD4+ T-cell proliferation and IFN- γ production in response to polyclonal activators is inhibited by *B. pseudomallei*-infected neutrophils via Programmed death ligand 1 (PD-L1) [14]. Overall, for generation of adaptive immune responses to *B. pseudomallei* excessive neutrophil activation and *B. pseudomallei* invasion of neutrophils may be detrimental to the optimal priming environment.

This study shows people with diabetes and acute melioidosis have a depressed cellular response to the bacteria than people without diabetes. This stunted cellular response is of interest because of the increased risk of melioidosis in people with diabetes. The mechanisms of immune dysregulation to intracellular pathogens seen in diabetes are an ongoing focus of research [43]. There are parallels to the relationship between adaptive immunity to tuberculosis and diabetes [44]. In the diabetic mouse model of aerosol challenge with tuberculosis [45], lower IFN- γ levels in the lung were found early in the infection (2 weeks post challenge) compared to non-diabetic mice, but the difference in IFN- γ levels had disappeared by 4 weeks. A temporal delay in the adaptive cellular response to TB was associated with poor early control of intracellular bacteria and ultimately a higher bacterial burden and more inflammation [46]. Our data support a similar relationship in humans and *B. pseudomallei*.

Although patients with diabetes had lower cellular responses to *B. pseudomallei* while acutely unwell, and lower cellular responses were associated with increased mortality, patients with diabetes did not have an increased mortality in this study, and diabetes was not an independent predictor of death in the multivariable logistic regression model. Previous studies have reported a survival benefit from the diabetes drug glyburide [47] but only one patient in this study took glyburide and use of any sulphonylurea drug was not associated with mortality. Elucidation of the susceptibility of diabetics to this bacterium will have wider implications for prevention of infections in diabetics and those with immune failure.

Patients with melioidosis and underlying known diabetes have poor glycaemic control compared to diabetic out-patients in the region. Healthy patients attending a diabetes outpatient clinic are unlikely to represent the same population because these are people actively engaging with management of their diabetes by attending clinic, whilst the patients in the *Melioid Cohort* have come to medical services because of acute illness. Nevertheless the fact that all but seven patients in the *Melioid Cohort* population with pre-diagnosed diabetes were receiving drug therapy, including insulin in the majority of cases, is evidence of receiving diabetes care. The interaction between diabetes and tuberculosis, another intracellular pathogen, is well known [48], and tuberculosis is reported to worsen glycaemic control [49]. We hypothesise that chronic exposure to *B. pseudomallei* may worsen glycaemic control. Because of the impact of rising prevalence of diabetes in Thailand [5] on morbidly and mortality, exacerbation by wide-spread *B. pseudomallei* exposure would have major public health implications. Longitudinal study of the relationship between *B. pseudomallei* presence in fields, human infection and glycaemic control are warranted.

Conclusions

This is the first study to measure the kinetics of a cellular immune response in from acute disease to long-term follow-up, and link results to outcome. Lower cellular responses are associated with increased mortality, supporting a pivotal role for cellular immunity in controlling this bacterium. The development of vaccine strategies that seek to enhance both cell mediated and humoral immunity may be beneficial for successful control of this disease.

Supporting Information

S1 Fig. Relationship between neutrophil count on admission (Week 0) and cellular response to B. pseudomallei on admission, 12 and 52 weeks later. The neutrophil count in cells per microliter for patients acutely unwell with melioidosis (*Melioid Cohort Week 0*) showed a negative correlation with the IFN- γ ELIspot response to *B. pseudomallei* (in spot forming cells per million peripheral blood mononuclear cells = SFC/10⁶ PBMC) for the same patients on admission (Week 0) and following recovery 12 and 52 weeks later, calculated by Spearman's rank correlation coefficient. (TIF)

Acknowledgments

We are grateful to the patients and staff of Sappasithiprasong Hospital. The authors wish to thank Brian Angus, Stuart Blacksell and Gumphol Wongsuvan for support establishing the study; Adrian Hill, Daniel Altmann, Rosemary Boyton and Jeanne Salje for helpful discussion; Daniel Paris for use of the ELIspot reader; Jeremy Chalk and Prapass Wannapinij for the study database and Nicholas White for overall support of the study.

Author Contributions

Conceived and designed the experiments: HAF DL SJD. Performed the experiments: KJ SC MS PA MH PS. Analyzed the data: KJ MS DL SJD. Contributed reagents/materials/analysis tools: NC VW PT NPJD. Wrote the paper: KJ MS NC VW HAF DL SJD.

References

- 1. Cheng AC, Currie BJ (2005) Melioidosis: epidemiology, pathophysiology, and management. Clin Microbiol Rev 18: 383–416. PMID: <u>15831829</u>
- Chierakul W, Rajanuwong A, Wuthiekanun V, Teerawattanasook N, Gasiprong M, et al. (2004) The changing pattern of bloodstream infections associated with the rise in HIV prevalence in northeastern Thailand. Transactions of the Royal Society of Tropical Medicine and Hygiene 98: 678–686. PMID: <u>15363648</u>
- Limmathurotsakul D, Wongratanacheewin S, Teerawattanasook N, Wongsuvan G, Chaisuksant S, et al. (2010) Increasing incidence of human melioidosis in Northeast Thailand. Am J Trop Med Hyg 82: 1113–1117. doi: <u>10.4269/ajtmh.2010.10-0038</u> PMID: <u>20519609</u>

- Currie BJ, Dance DA, Cheng AC (2008) The global distribution of Burkholderia pseudomallei and melioidosis: an update. Trans R Soc Trop Med Hyg 102 Suppl 1: S1–4. doi: <u>10.1016/S0035-9203(08)</u> <u>70002-6</u> PMID: <u>19121666</u>
- Deerochanawong C, Ferrario A (2013) Diabetes management in Thailand: a literature review of the burden, costs, and outcomes. Global Health 9: 11. doi: 10.1186/1744-8603-9-11 PMID: 23497447
- UNFPA (2006) Population Ageing in East and South-East Asia: Current Situation and Emerging Challenges. Bangkok, Thailand: United Nations Population Fund; http://www.academia.edu/8844277/.
- Peacock SJ, Limmathurotsakul D, Lubell Y, Koh GC, White LJ, et al. (2012) Melioidosis vaccines: a systematic review and appraisal of the potential to exploit biodefense vaccines for public health purposes. PLoS Negl Trop Dis 6: e1488. doi: 10.1371/journal.pntd.0001488 PMID: 22303489
- Wiersinga WJ, Wieland CW, Dessing MC, Chantratita N, Cheng AC, et al. (2007) Toll-like receptor 2 impairs host defense in gram-negative sepsis caused by Burkholderia pseudomallei (Melioidosis). PLoS Med 4: e248. PMID: 17676990
- West TE, Chierakul W, Chantratita N, Limmathurotsakul D, Wuthiekanun V, et al. (2012) Toll-like receptor 4 region genetic variants are associated with susceptibility to melioidosis. Genes Immun 13: 38–46. doi: 10.1038/gene.2011.49 PMID: 21776015
- West TE, Chantratita N, Chierakul W, Limmathurotsakul D, Wuthiekanun V, et al. (2013) Impaired TLR5 functionality is associated with survival in melioidosis. J Immunol 190: 3373–3379. doi: <u>10.4049/</u> jimmunol.1202974 PMID: <u>23447684</u>
- Myers ND, Chantratita N, Berrington WR, Chierakul W, Limmathurotsakul D, et al. (2014) The role of NOD2 in murine and human melioidosis. J Immunol 192: 300–307. doi: <u>10.4049/jimmunol.1301436</u> PMID: <u>24298015</u>
- Hodgson KA, Morris JL, Feterl ML, Govan BL, Ketheesan N (2011) Altered macrophage function is associated with severe Burkholderia pseudomallei infection in a murine model of type 2 diabetes. Microbes Infect 13: 1177–1184. doi: 10.1016/j.micinf.2011.07.008 PMID: 21835260
- Chanchamroen S, Kewcharoenwong C, Susaengrat W, Ato M, Lertmemongkolchai G (2009) Human polymorphonuclear neutrophil responses to Burkholderia pseudomallei in healthy and diabetic subjects. Infect Immun 77: 456–463. doi: 10.1128/IAI.00503-08 PMID: 18955471
- Buddhisa S, Rinchai D, Ato M, Bancroft GJ, Lertmemongkolchai G (2015) Programmed Death Ligand 1 on Burkholderia pseudomallei-Infected Human Polymorphonuclear Neutrophils Impairs T Cell Functions. J Immunol 194: 4413–4421. doi: 10.4049/jimmunol.1402417 PMID: 25801435
- Kanaphun P, Thirawattanasuk N, Suputtamongkol Y, Naigowit P, Dance DA, et al. (1993) Serology and carriage of Pseudomonas pseudomallei: a prospective study in 1000 hospitalized children in northeast Thailand. J Infect Dis 167: 230–233. PMID: <u>7678106</u>
- Wuthiekanun V, Chierakul W, Langa S, Chaowagul W, Panpitpat C, et al. (2006) Development of antibodies to Burkholderia pseudomallei during childhood in melioidosis-endemic northeast Thailand. The American journal of tropical medicine and hygiene 74: 1074–1075. PMID: 16760522
- Jones AL, Beveridge TJ, Woods DE (1996) Intracellular survival of Burkholderia pseudomallei. Infect Immun 64: 782–790. PMID: 8641782
- Stevenson CR, Critchley JA, Forouhi NG, Roglic G, Williams BG, et al. (2007) Diabetes and the risk of tuberculosis: a neglected threat to public health? Chronic Illn 3: 228–245. PMID: 18083679
- Koh GC, Schreiber MF, Bautista R, Maude RR, Dunachie S, et al. (2013) Host responses to melioidosis and tuberculosis are both dominated by interferon-mediated signaling. PloS one 8: e54961. doi: <u>10.</u> <u>1371/journal.pone.0054961</u> PMID: <u>23383015</u>
- Dharakul T, Vejbaesya S, Chaowagul W, Luangtrakool P, Stephens HA, et al. (1998) HLA-DR and -DQ associations with melioidosis. Human immunology 59: 580–586. PMID: <u>9757914</u>
- Santanirand P, Harley VS, Dance DA, Drasar BS, Bancroft GJ (1999) Obligatory role of gamma interferon for host survival in a murine model of infection with Burkholderia pseudomallei. Infect Immun 67: 3593–3600. PMID: 10377144
- Haque A, Easton A, Smith D, O'Garra A, Van Rooijen N, et al. (2006) Role of T cells in innate and adaptive immunity against murine Burkholderia pseudomallei infection. The Journal of infectious diseases 193: 370–379. PMID: 16388484
- Brown AE, Dance DA, Suputtamongkol Y, Chaowagul W, Kongchareon S, et al. (1991) Immune cell activation in melioidosis: increased serum levels of interferon-gamma and soluble interleukin-2 receptors without change in soluble CD8 protein. J Infect Dis 163: 1145–1148. PMID: 1902247
- Lauw FN, Simpson AJ, Prins JM, Smith MD, Kurimoto M, et al. (1999) Elevated plasma concentrations of interferon (IFN)-gamma and the IFN-gamma-inducing cytokines interleukin (IL)-18, IL-12, and IL-15 in severe melioidosis. J Infect Dis 180: 1878–1885. PMID: <u>10558944</u>

- Lauw FN, Simpson AJ, Prins JM, van Deventer SJ, Chaowagul W, et al. (2000) The CXC chemokines gamma interferon (IFN-gamma)-inducible protein 10 and monokine induced by IFN-gamma are released during severe melioidosis. Infection and immunity 68: 3888–3893. PMID: <u>10858199</u>
- Ketheesan N, Barnes JL, Ulett GC, VanGessel HJ, Norton RE, et al. (2002) Demonstration of a cellmediated immune response in melioidosis. J Infect Dis 186: 286–289. PMID: <u>12134268</u>
- Tippayawat P, Saenwongsa W, Mahawantung J, Suwannasaen D, Chetchotisakd P, et al. (2009) Phenotypic and functional characterization of human memory T cell responses to Burkholderia pseudomallei. PLoS neglected tropical diseases 3: e407. doi: <u>10.1371/journal.pntd.0000407</u> PMID: <u>19352426</u>
- Limmathurotsakul D, Funnell SG, Torres AG, Morici LA, Brett PJ, et al. (2015) Consensus on the Development of Vaccines against Naturally Acquired Melioidosis. Emerg Infect Dis 21.
- Wuthiekanun V, Langa S, Swaddiwudhipong W, Jedsadapanpong W, Kaengnet Y, et al. (2006) Short report: Melioidosis in Myanmar: forgotten but not gone? Am J Trop Med Hyg 75: 945–946. PMID: <u>17123993</u>
- Ashdown LR, Guard RW (1984) The prevalence of human melioidosis in Northern Queensland. The American journal of tropical medicine and hygiene 33: 474–478. PMID: <u>6731680</u>
- (2009) International Expert Committee report on the role of the A1C assay in the diagnosis of diabetes. Diabetes Care 32: 1327–1334. doi: 10.2337/dc09-9033 PMID: 19502545
- Bursac Z, Gauss CH, Williams DK, Hosmer DW (2008) Purposeful selection of variables in logistic regression. Source Code Biol Med 3: 17. doi: <u>10.1186/1751-0473-3-17</u> PMID: <u>19087314</u>
- Walsh AL, Smith MD, Wuthiekanun V, Suputtamongkol Y, Chaowagul W, et al. (1995) Prognostic significance of quantitative bacteremia in septicemic melioidosis. Clin Infect Dis 21: 1498–1500. PMID: 8749644
- Currie BJ, Ward L, Cheng AC (2010) The epidemiology and clinical spectrum of melioidosis: 540 cases from the 20 year Darwin prospective study. PLoS Negl Trop Dis 4: e900. doi: <u>10.1371/journal.pntd.</u> 0000900 PMID: <u>21152057</u>
- 35. Haque A, Chu K, Easton A, Stevens MP, Galyov EE, et al. (2006) A live experimental vaccine against Burkholderia pseudomallei elicits CD4+ T cell-mediated immunity, priming T cells specific for 2 type III secretion system proteins. The Journal of infectious diseases 194: 1241–1248. PMID: 17041850
- Lertmemongkolchai G, Cai G, Hunter CA, Bancroft GJ (2001) Bystander activation of CD8+ T cells contributes to the rapid production of IFN-gamma in response to bacterial pathogens. Journal of immunology 166: 1097–1105.
- Ye Z, Lee CM, Sun GW, Gan YH (2008) Burkholderia pseudomallei infection of T cells leads to T-cell costimulation partially provided by flagellin. Infect Immun 76: 2541–2550. doi: <u>10.1128/IAI.01310-07</u> PMID: <u>18347031</u>
- Brown KA, Brain SD, Pearson JD, Edgeworth JD, Lewis SM, et al. (2006) Neutrophils in development of multiple organ failure in sepsis. Lancet 368: 157–169. PMID: <u>16829300</u>
- Laws TR, Smither SJ, Lukaszewski RA, Atkins HS (2011) Neutrophils are the predominant cell-type to associate with Burkholderia pseudomallei in a BALB/c mouse model of respiratory melioidosis. Microb Pathog 51: 471–475. doi: 10.1016/j.micpath.2011.07.002 PMID: 21798336
- Easton A, Haque A, Chu K, Lukaszewski R, Bancroft GJ (2007) A critical role for neutrophils in resistance to experimental infection with Burkholderia pseudomallei. The Journal of infectious diseases 195: 99–107. PMID: <u>17152013</u>
- Pillay J, Kamp VM, van Hoffen E, Visser T, Tak T, et al. (2012) A subset of neutrophils in human systemic inflammation inhibits T cell responses through Mac-1. J Clin Invest 122: 327–336. doi: <u>10.1172/</u> <u>JCI57990</u> PMID: <u>22156198</u>
- Gabrilovich DI, Nagaraj S (2009) Myeloid-derived suppressor cells as regulators of the immune system. Nat Rev Immunol 9: 162–174. doi: <u>10.1038/nri2506</u> PMID: <u>19197294</u>
- Hodgson K, Morris J, Bridson T, Govan B, Rush C, et al. (2015) Immunological mechanisms contributing to the double burden of diabetes and intracellular bacterial infections. Immunology 144: 171–185. doi: 10.1111/imm.12394 PMID: 25262977
- Martinez N, Komfeld H (2014) Diabetes and immunity to tuberculosis. Eur J Immunol 44: 617–626. doi: 10.1002/eji.201344301 PMID: 24448841
- Martens GW, Arikan MC, Lee J, Ren F, Greiner D, et al. (2007) Tuberculosis susceptibility of diabetic mice. Am J Respir Cell Mol Biol 37: 518–524. PMID: <u>17585110</u>
- 46. Vallerskog T, Martens GW, Kornfeld H (2010) Diabetic mice display a delayed adaptive immune response to Mycobacterium tuberculosis. J Immunol 184: 6275–6282. doi: <u>10.4049/jimmunol.1000304</u> PMID: 20421645

- Koh GC, Maude RR, Schreiber MF, Limmathurotsakul D, Wiersinga WJ, et al. (2011) Glyburide is antiinflammatory and associated with reduced mortality in melioidosis. Clin Infect Dis 52: 717–725. doi: <u>10.</u> <u>1093/cid/ciq192</u> PMID: <u>21293047</u>
- **48.** Bridson TL, Govan BL, Norton RE, Schofield L, Ketheesan N (2014) The double burden: a new-age pandemic meets an ancient infection. Trans R Soc Trop Med Hyg.
- Lonnroth K, Roglic G, Harries AD (2014) Improving tuberculosis prevention and care through addressing the global diabetes epidemic: from evidence to policy and practice. Lancet Diabetes Endocrinol 2: 730–739. doi: <u>10.1016/S2213-8587(14)70109-3</u> PMID: <u>25194886</u>